Author’s response to reviews

Title: Mortality, cause of death and risk factors in patients with alcohol use disorder alone or poly-substance use disorders: a 19-year prospective cohort study

Authors:

Arne Jan Hjemsæter (Arne.Jan.Hjemsaeter@sykehuset-innlandet.no)

Jørgen Bramness (j.g.bramness@medisin.uio.no)

Robert Drake (RobertDrake@westat.com)

Ivar Skeie (ivar.skeie@gmail.com)

Bent Monsbakken (bent.monsbakken@gmail.com)

Jurate Benth (jurate.saltyte-benth@medisin.uio.no)

Anne Landheim (Anne.Signe.Landheim@sykehuset-innlandet.no)

Version: 2 Date: 11 Mar 2019

Author’s response to reviews:

Thank you to both reviewers for informed and detailed reviews, and editor for the last revision. In this revision no suggestions from the reviewers were addressed to us. We have gone through all four suggestions by the editor and made changes accordingly.